Author's response to reviews

Title: Invasive pneumococcal infections in Vellore, India: Clinical characteristics and distribution of serotypes

Authors:

Camilla Elisson (elisson.camilla@gmail.com)
Viktor Molander (viktormolander@hotmail.com)
Veeraraghavan Balaji (veeraraghavan.balaji@facebook.com)
Erik Backhaus (erik.backhaus@vgregion.se)
James John (james_cmc2002@yahoo.co.in)
Rosemol Vargheese (rosemol.vargheese@facebook.com)
Ranjith Jayaraman (ranjithcmc@gmail.com)
Rune Andersson (rune.andersson@gu.se)

Version: 2 Date: 25 July 2013

Author's response to reviews: see over
Dear Editor,

We have now addressed the comments we received after the last submission of our manuscript. All changes are listed/explained below.

All the best,

Camilla Elisson, Viktor Molander and co-authors
Gothenburg/Stockholm, Sweden, 130725

1. Corrected the english language throughout the entire paper using Ednaz Group service that You recommended.
2. Added ”background” to the abstract.
3. Changed the format of the references accorting to the guidelines. Also, we double-checked all references and changed no 15 that was accidentally wrong.
4. Changed table and figure titles/legends according to guidelines.